Author's response to reviews

Title: Determining the Validity and Reliability of the Chinese Version of the Kidney Disease Quality of Life Questionnaire (KDQOL-36TM)

Authors:

   Xingjuan Tao (tao.xingjuan@connect.polyu.hk)
   Susan Ka Yee Chow Dr (susan.chow@polyu.edu.hk)
   Frances KY Wong Prof (frances.wong@polyu.edu.hk)

Version: 5 Date: 30 May 2014

Author's response to reviews:

Dear reviewers,

Thank you very much for your constructive comments to improve the overall quality of the paper. The authors carefully revised the manuscript I hope the revisions are to your satisfaction.

Regards,
Susan